C++ Programming OODS1 Tutorial 1

© 2000, Uwe Rastofer UniversitätErlangen Nürnberg,Informatik4

Reproduktion oder Verwendung dieser Unterlage zu Lehrzwecken außerhalb der Universität Erlangen ist ohne Genehmigung der Autoren unzulässig

Guidelines for OODS C++ Assignments

a) ForallC++-relatedassignments use a veryrecent GNUC++Compiler(also knownas g++).Atested version is installedin /local/gcc-2.95.2/bin forthe Solaris and Linuxmachines. We tested the compilerso that you can be sure that allfeatures you need forthe assignments willworkas expected. Other compilers,likeolder versions of GCC or the Sun Workshop5.0 Compilersin /opt/SUNWspro/bin, either don’t have the features at all or the implementations are faulty.

b) For every class you write, there must be two files. One header file containing the declarationofthe interface(and maybe some inlinemethods) and one implementation filecontainingthe method bodies. E.g. fora class Person the header fileshould be called Person.h and the implementation filePerson.cpp. Inadditionthere must be one filemain.cpp whichcontains the main() function.This function is used to test your classes.

c) Everyclass must be separately compiled(g++ -pedantic -c <class>.cpp) and afterwards all parts are linked together (g++ -o <executable> <class1>.o <class2>.o ...). The preferred way to do this is to create a Makefile and use the

make tool.The executable willcomplainabout a missinglibrarynamed libstdc++.

Thislibrarycontains the runtimecode forANSIC++.Whenlinkingthe executable you have to give an additional parameter -R/local/gcc-2.95.2/lib/gcc-lib/ <arch>/2.95.2 where <arch> is sparc-sun-solaris2.6 forSolaris and i686-

pc-linux-gnu forLinux.Thistellsthe linkertoremember the place wherethislibrary

is stored. d) Mail questions and comments torastofer@informatik.uni-erlangen.de.

C++ Programming OODS1 Tutorial 2

© 2000, Uwe Rastofer UniversitätErlangen Nürnberg,Informatik4

Reproduktion oder Verwendung dieser Unterlage zu Lehrzwecken außerhalb der Universität Erlangen ist ohne Genehmigung der Autoren unzulässig

Homework Assignment #8: Simple table (Due Dec. 21, 2000 9:30) 0 points

This assignment is designed to get you accustomed to the C++environment.Youhave to writea simple Table class whichstores Person objects. The internalrepresentation should be an array. The array size is givenwhen creating the table. Youwillfinda simple implementation of thePerson class in /proj/i4oods/pub/assignment8. a) Implement a constructor which takes the table size as an argument. Dynamically allocate the storage for the table. Remember to free the storage when the table is destroyed. How can you modifythe constructor so that the table size is no longer mandatory (Hint: default size)?

b) Writea print method whichtakes an outputstream as an argument (void print(

ostream& ) and outputs the table one entry in a line. Printingdoesn’tmodifythe

table - how do you specify this in the method declaration? c) Writeinsert and remove methods to modifythe table. They both should return an

int to show if the operation was successful.

d) Writea lookup method to findthe firstperson witha matchingname. (Hint:Lookups don’t change the table either).

e) Writea copy constructor(Table(const Table&)) whichtakes another table as an argument and initializes the new table with the data of the other one.

f) Writean equals method that checks if two tables have equal entries. Two tables should be equal if they contain the same persons (in any sequence).

Hint: Check your implementation that it does not leave any garbage objects behind.